Author’s response to reviews

Title: MDRD vs. CKD-EPI in comparison to 51Chromium EDTA: A cross sectional study of Malaysian CKD cohort

Authors:
Maisarah Jalalonmuhali (mai_jalal@yahoo.com)
Soo Kun Lim (sookun_73@yahoo.com)
Mohammad Nazri Md Shah (mnazri@ummc.edu.my)
Kok Peng Ng (kokpeng5@yahoo.com)

Version: 2 Date: 31 Oct 2017

Author’s response to reviews:

Reviewers response written in the attached cover letter.